Author's response to reviews

Title: Intraprostatic Botulinum Toxin Type A injection in patients with benign prostatic enlargement: Duration of the effect of a single injection

Authors:

João FA Silva (jfalturas@gmail.com)
Rui Pinto (ruipinto@mac.com)
Tiago G Carvalho (tgorgal@sapo.pt)
Francisco Botelho (francisco.botelho@sapo.pt)
Pedro N Silva (pedrocdns@hotmail.com)
Rui Oliveira (ruisoliveira@gmail.com)
Carlos Silva (carsil@med.up.pt)
Francisco R Cruz (cruzfmr@med.up.pt)
Paulo O Dinis (paulodinisoliveira@gmail.com)

Version: 3 Date: 17 July 2009

Author's response to reviews: see over
Dear Biomed Central Editorial Team

Regarding our recently submitted manuscript “Intraprostatic Botulinum Toxin Type A injection in patients with benign prostatic enlargement: Duration of the effect of a single injection” we are sending a second revised version of the manuscript which I hope will address the reviewers' comments. We thank the reviewers for their critique of our work. Their suggestions helped us to improve our paper.

The revisions included in the manuscript and responses to issues raised by the reviewers are as follows:

Reviewer 2: Rahmi Onur

On page 4, first sentence still needs to be revised. “Man” should be “man” and “.....has been investigated in most recent years as a treatment for multiple disorders, including lower urinary tract dysfunction” should be corrected to ““..recently, ....has been investigated in the treatment of multiple disorders, including lower urinary tract dysfunction”

- Done
Page 5, last sentence of introduction. “In addition the evolution of total serum PSA, lower urinary tract symptoms and urinary.......”, word evolution should be corrected to evaluation.

- Done

- References and manuscript were overviewed.

We hope that our revised manuscript is suitable for publication

Sincerely

Francisco Cruz, MD, PhD

Department of Urology

Hospital S. Joao - Porto

Portugal